Author's response to reviews

Title: A three-group study, internet-based, face-to-face based and standard-management after acute whiplash associated disorders (WAD) - choosing the most efficient and cost-effective treatment: study protocol of a randomized controlled trial.

Authors:

Anne Söderlund (anne.soderlund@mdh.se)
Annika Bring (annika.bring@neuro.uu.se)
Pernilla Åsenlöf (pernilla.asenlof@neuro.uu.se)

Version: 2 Date: 17 July 2009

Author's response to reviews: see over
Dear Editor

We have today re-submitted an original manuscript of a study protocol by Anne Söderlund, Professor, PhD, Annika Bring, Doctoral student, RPT and Pernilla Åsenlöf, PhD, titled: “A three-group study, internet-based, face-to-face based and standard-management after acute whiplash associated disorders (WAD) – choosing the most efficient and cost-effective treatment: study protocol of a randomized controlled trial.” for a possible publication in the BMC Musculoskeletal Disorders.

We have made the following revisions. Major revisions:
The job titles and qualifications have been removed from the list of authors on the title page. Reference 3 is now accurate. Reference 43 is now accurate. Reference 51 is now accurate. The image file for the figure no longer includes the title or figure number. Additional files are listed properly in the end of the paper as suggested. Additional files 1 and 2 are mentioned in the text instead of Table 1 or Figure 1.

Minor revisions:
The title page is properly organized. Funding has been moved to the Acknowledgements section. Keywords have been removed. Introduction – is renamed to Background. Methods - is rename to Methods/design. Table and figure placement texts have been removed. Figure cropping – unfortunately we did not manage to crop the figure. The reference [17] is ok now. The (Söderlund, 2002 #11[20]) reference is ok now.

We hope that these changes are satisfying. We are looking forward to hear from you in the near future.

Yours sincerely

On behalf and by permission of the other authors: Anne Söderlund, Professor, Ph.D.

Corresponding author: Anne Söderlund, Professor, PhD, Department of Physiotherapy, School of Health, Care and Social Welfare, Mälardalen University, Västerås, Sweden and Section of Physiotherapy, Department of Neuroscience, Uppsala University, Sweden;